Author's response to reviews

Title: Educational and labor markets of dentists in Mexico. Elements for its strategic planning.

Authors:

Luz Maria LM Gonzalez (luz.gonzalez@insp.mx)
Gustavo G Nigenda (gnigenda@insp.mx)
Maria Cecilia MC Gonzalez (cecilia.gonzalez@insp.mx)

Version: 3 Date: 21 December 2011

Author's response to reviews:

Cuernavaca, Mexico, December 20, 2011.

Editorial Board
Human Resources for Health

Dear Editors:

We are sending attached the new version of the document Dentist education and labor markets in Mexico. Elements for strategic planning. This version contains the latest changes suggested by the reviewers and received by us on December 13th.

The changes are the following:

a) On page 7, paragraph 2, line 2, the words “massively participate” was left out. It was replaced by the phrase “88.2% were practicing ….”

b) Page 7, paragraph 3, line 11, the number 1,000,000 was replaced by 10,000.

We want to express our gratitude for the time taken by the authors to read and comment on the previous version. We hope that responses are satisfactory and the document could be published in your prestige journal.

With best regards,

The authors.